Author's response to reviews

Title: High dose intravenous iron, mineral homeostasis and intact FGF23 in normal and uremic rats

Authors:

Eva Gravesen (eva.gravesen@regionh.dk)
Jacob Hofman-Bang (jacob.hofman-bang@regionh.dk)
Maria L Mace (marialerchenielsen@gmail.com)
Ewa Lewin (ewa.lewin@regionh.dk)
Klaus Olgaard (olgaard@rh.dk)

Version: 2 Date: 22 October 2013

Author's response to reviews:

Dear Editor,

The three Figures have been cropped according to your request.

Yours sincerely

Klaus Olgaard, M.D.
Professor, dr. med.
University of Copenhagen
Nephrological Department P 2132
Rigshospitalet
9 Blegdamsvej
DK 2100 Copenhagen
Denmark.
Phone: +45 3545 2130
Fax: +45 3545 2672
E-mail: olgaard@rh.dk
E-mail 2: olgaard@dadlnet.dk
Mobil phone: +45 4081 1722